\anrdoc{1 page maximum par personne (Cf. 5.3).}

\newenvironment{cvenv}[4]{%
  % #1 : {Nom}{Prenom}{age}
  % #2 : situation actuelle
  % #3 : cursus
  % #4 : {#}{recent pub}
  \noindent
  \begin{tabular}{|p{3.3cm}|p{4.3cm}|p{1.4cm}|p{6.3cm}|}\hline
    \def\tmp##1##2##3{Name: ##1&First Name: ##2&age: ##3}\tmp#1
    &
    \def\tmp##1##2{##1 publications: ##2}\tmp#4
    \\\hline
  \end{tabular}
  \begin{description}
    \item[Situation] #2
    \item[Cursus] #3
  }{\end{description}\vspace*{1.ex}}

\begin{cvenv}
  {{Greiner}{Alain}{58}}
  {Professeur UMPC}
  {HDR Thesis (1982)}
  {{100}{\cite{ag-1} \cite{ag-2} \cite{ag-3} \cite{ag-4}}}
  
  \item[Projects]\mbox{}
    \begin{itemize}
      \item Work on TSAR CATRENE project \cite{disydent05}.
      \item Worked on SoCLib ANR project\cite{disydent05}.
      \item Worked on COSY european project \cite{cosy}.
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Aug\'e}{Ivan}{50}}
  {Ma\^itre de conf\'erences at ENSIIE (Evry Essonne)}
  {HDR Thesis (2008), Thesis (1990), ENSIIE Engineer (1983)}
  {{20}{\cite{ia-hdr} \cite{ugh08} \cite{disydent05}}}
  
  \item[Courses of lecture] System, Kernel, Network, Computer Architecture, Programing.
  \item[Projects]\mbox{}
    \begin{itemize}
      \item Worked on COSY european project \cite{cosy}.
      \item Work   on UGH  since 1996 \cite{ugh08}.
      \item Worked on Disydent \cite{disydent05}.
    \end{itemize}
  \item[Other work experiences]\mbox{}
    \begin{itemize}
      \item Work as system administrator since six years at ENSIIE.
      \item Worked as a senior engineer for four years at Philips.
      \item Worked as engineer for two years at ANIRAL-UTEC.
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Nguyen}{Huy-Nam}{56}}
  {Head of MVS Departement at Bull S.A.S}
  {Thesis (1982), ENSMP Engineer (1977)}
  {{5}{}}
  \item[Projects]\mbox{}
    \begin{itemize}
      \item Coordinator of TSAR CATRENE project \cite{disydent05}.
      \item Worked on projects: ITEA/Care4Me, Medea+/NEVA, Medea/MESA, Medea/AT403, Medea/AT407, Jessi/AC3, Jessi/AC8
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Feautrier}{Paul}{71}}
  {Emeritus Professor}
  {Th\`ese d'Etat, 1968}
  {{100}{\cite{Feau:92aa} \cite{Feau:92bb} \cite{FP:96} \cite{Feau:96} 
       \cite{syntol}}}
  \item[Course of Lectures]\mbox{}
       System, System Kernel, Automatic Parallelization, Operation Research,
       Compilation (at Universit\'e P. et M. Curie, Universit\'e de Verasilles
       and Ecole Normale Sup\'erieure de Lyon).
  \item[Projects]\mbox{}
    \begin{itemize}
       \item ITEA MARTES, 2007--2009.
    \end{itemize}
  \item[Other Work Experience]\mbox{}
    \begin{itemize}
       \item PhD thesis in Computational Astrophysics (HPC) 1960--1968.
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Alias}{Christophe}{33}}
  {INRIA researcher (CR2)}
  {PhD (2005)}
  {{11}{\cite{bee}} \cite{ca:nuca} \cite{ca:chuba} \cite{ca:arc}}
  \item[Course of Lectures]\mbox{}
       Compilers, programming languages, computer architecture, networks.
\end{cvenv}
%
\begin{cvenv}
  {{P\'etrot}{Fr\'ed\'eric}{44}}
  {Professor }
  {PhD (1994), HDR (2003)}
  {{75}{\cite{disydent05} \cite{JerrayaPetrot} \cite{mutek} \cite{dna}}}
  \item[Course of Lectures]\mbox{}
       Computer Architecture, programming language, operating systems,
       practical compilation, high-level synthesis.
  \item[Projects]\mbox{}
    \begin{itemize}
       \item Hospi, SoCLib (ANR)
       \item Medea AT403, Lomosa
       \item OpenTLM, Sceptre (FUI)
       \item COSY, Sprint (ESPRIT)
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Muller}{Olivier}{30}}
  {Associate Professor}
  {PhD (2007)}
  {{10}{}}
  \item[Course of Lectures]\mbox{}
       Computer Architecture, assembly language programming, FPGA
  \item[Projects]\mbox{}
    \begin{itemize}
       \item Medea iGlance
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Charot}{François}{52}}
  {INRIA researcher }
  {PhD (1982)}
  {{50}{\cite{Martin09c} \cite{Martin09d} \cite{Wolinski09a} \cite{RAFFIN:2010:INRIA-00539874:1}}}
  \item[Course of Lectures]\mbox{}
       embedded system design
 \item[Projects]\mbox{}
    \begin{itemize}
       \item SocLib and ROMA ANR projects
%       \item Nano2012 RecMotifs project
   \end{itemize}
\end{cvenv}
%
\begin{cvenv}
{{Derrien}{Steven}{36}}
{Associate Professor}
{PhD (2002)}
{{20}{\cite{sd:lomita}} \cite{sd:mmalpha} \cite{sd:hmmer} \cite{sd:rdisk}}
\item[Course of Lectures]\mbox{}
Principles of operating systems, Design of embedded systems, programming languages.
\item[Projects]\mbox{}
\begin{itemize}
\item ANR-Biowic, 2009--
%\item Nano2012 S2S4HLS, 2009--
\end{itemize}
\end{cvenv} 
%
\begin{cvenv}
  {{Coussy}{Philippe}{36}}
  {Associate Professor}
  {PhD (2003)}
  {{47}{ \cite{5605303} \cite{gaut08} \cite{HLSBOOK} \cite{DBLP:journals:dt:CoussyGMT09}
         \cite{DBLP:journals:vlsisp:ThabetCHM09}}}
  \item[Course of Lectures]\mbox{}
       Computer Architecture, programming language, operating systems,
       SoC Design, high-level synthesis.
  \item[Projects]\mbox{}
    \begin{itemize}
       \item SoCKeT (FUI)
       \item SoCLib (ANR)
       \item SystemC'Mantic (RNTL)
       \item FLASH (ARC INRIA)
       \item ALIPTA (RNRT)
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Chavet}{Cyrille}{32}}
  {Associate Professor}
  {PhD (2007)}
{{20}{\cite{CHAVET:2007:HAL-00153994:1} \cite{DBLP:conf:iccad:ChavetACCJUM07}
\cite{DBLP:conf:glvlsi:ChavetCUM07} \cite{5605303}}}
  \item[Course of Lectures]\mbox{}
       Parrallel programming, VHDL, real time programming, database networking 
  \item[Projects]\mbox{}
    \begin{itemize}
       \item SoCKeT (FUI)
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Heller}{Dominique}{44}}
  {Research Engineer}
  {PhD (1998)}
  {{24}}
  \item[Course of Lectures]\mbox{}
       FPGA, VHDL, programming language, HLS 
  \item[Projects]\mbox{}
    \begin{itemize}
       \item SoCKeT (FUI)
       \item SoCLib (ANR)
       \item SystemC'Mantic (RNTL)
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Lemonnier}{Fabrice}{43}}
  {Research program manager}
  {PhD (1996), Master in parallel computing (1992), EFREI Engineer (1991)}
  {{3}{}}
  \item[Projects]\mbox{}
    \begin{itemize}
       \item Coordinator of Teraops project (p\^ole de comp\'etitivit\'e System\verb+@+tic) to develop a heterogeneous manycore
       \item Works on FOSFOR and PREIA ANR project on reconfigurable computing, and on CHAPI (p\^ole de comp\'etitivit\'e System\verb+@+tic)
    \end{itemize}
\end{cvenv}
%
\begin{cvenv}
  {{Brelet}{Paul}{28}}
  {software developer}
  {EFREI Engineer (2007)}
  {{4}{}}
  \item[Projects]\mbox{}
    \begin{itemize}
       \item Worked on toolset in MORPHEUS (ICT FP6)
       \item Works on toolset in SOCKET (p\^ole de comp\'etitivit\'e System\verb+@+tic), toolset in RTSimex (ANR)
    \end{itemize}
\end{cvenv}

